July 13, 2006

Dear editor of the BMC Medical Research Methodology,

Attached, please find the revised manuscript entitled: “Country development and manuscript selection bias: a review of published studies “which is corrected according to the formatting change requests:

- Qualifications - the authors' qualifications are removed
- Discussion - the Discussions title is changed to 'Discussions and Conclusions'.

Sincerely yours

Reza Yousefi-Nooraie, MD.

Center of Evidence Based Medicine, Tehran University of Medical Sciences, Iran

Address: P.O. Box: 13145-967, Tehran, Iran

Email: ryousefi@razi.tums.ac.ir

Telephone: ++98-21-6901838